Author's response to reviews

Title: Pragmatic trials in primary care Methodological challenges and solutions demonstrated by the DIAMOND-study

Authors:

Gerdine A.J. Fransen (gerdinefransen@hotmail.com)
Corine J. Van Marrewijk (c.vanmarrewijk@mdl.umcn.nl)
Suhretra Mujakovic (suhretra@backus.com)
Jean W.M. Muris (jean.muris@hag.unimaas.nl)
Robert J.F. Laheij (r.laheij@mdl.umcn.nl)
Mattijs E. Numans (m.e.numans@umcutrecht.nl)
Niek J. De Wit (n.j.dewit@umcutrecht.nl)
Melvin Samsom (m.samsom@umcutrecht.nl)
Jan B.M.J. Jansen (j.jansen@mdl.umcn.nl)
J. Andre Knottnerus (andre.knottnerus@hag.unimaas.nl)

Version: 4 Date: 17 April 2007

Author's response to reviews: see over
Dear editors,

Please find enclosed our revised manuscript entitled “Pragmatic trials in primary care. Methodological challenges and solutions demonstrated by the DIAMOND-study” for publication in BMC Medical Research Methodology.

The manuscript has been improved by making all formatting changes requested.

We certify that this work has not been published, simultaneously submitted, or already accepted for publication elsewhere and that submission for publication has been approved by all authors and by the departments where the work was carried out.

We are looking forward to hear from you.

Yours faithfully,

Gerdine Fransen
On behalf of all co-authors